Author’s response to reviews

Title: Acute and mixed alcohol intoxications in asylum seekers presenting to an urban emergency department in Switzerland

Authors:

Adam Brown (brownad@newschool.edu)
Martin Müller (martin.mueller2@insel.ch)
Trevor Hirschi (trevor.hirschi@insel.ch)
Jonathan Henssler (jonathan.henssler@charite.de)
Katharina Rönz (katharina.roenz@insel.ch)
Aristomenis Exadaktylos (aristomenis.exadaktylos@insel.ch)
David Srivastava (david.srivastava@insel.ch)

Version: 2 Date: 24 Apr 2019

Author’s response to reviews:

> Please include the email addresses of all the authors on the title page. Please ensure that it is identical to the one entered on the submission system.

This has been addressed. All of the emails now appear on the title page.

>> Please include an Abstract section heading.

This has been added. We have since added this section heading.

>> Please move the List of Abbreviations to after the Conclusions section.

This has been moved.

>> Please rename Author roles as Authors' Contribution. We suggest the following format (please use initials to refer to each author's contribution): AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the
immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination and helped to draft the manuscript. All authors read and approved the final manuscript.

We have renamed the section to Authors' Contribution. We have also revised the format to adhere to the one provided above.

Thank you very much.

Best,

Adam